\begin{table}%t5
\caption{\label{tab5}Separations and position angles for previously known multiple systems.}
\small%\centering
\par
\begin{tabular}{l r r r r}
 \hline\hline\noalign{\smallskip}
2MASS ID & $\rho$ & $\theta$ & Epoch & Ref.\\
  &	[$\arcsec$] & 	 [$\degr$]	&	&	\\
\hline
\par
J00503319+2449009	&	1.0	&	315	&	1960	&	1	\\
	&	2.080	&	316.0	&	1991.25	&	2	\\
	&	1.648	&	317.12	&	2002.64	&	3	\\
	&	1.305	&	318.9	&	2008.86	&	4	\\
\\
J01093874-0710497	&	2.680	&	77.0	&	1991.25	&	2	\\
	&	2.554	&	74.5	&	2008.87	&	4	\\
\\
J02165488-2322133	&	4.3	&	315	&	1998.67	&	5	\\
	&	4.369	&	314.1	&	2008.87	&	4	\\
\\
J04132663-0139211	&	0.79	&	217.11	&	1998.9	&	6	\\
	&	0.771	&	358.8/178.8	&	2008.88	&	4	\\
\\
J04373746-0229282	&	0.225	&	195	&	2003.05	&	7	\\
	&	0.093	&	189.5	&	2004.95	&	7	\\
	&	0.221	&	20.5	&	2008.88	&	4	\\
\\
J04441107-7019247	&	2.3	&	174	&	1990	&	8	\\
	&	2.654	&	157.5	&	2008.88	&	4	\\
\\
J05254166-0909123	&	0.537	&	69.40	&	2005.78	&	9	\\
	&	0.616	&	58.8	&	2008.86	&	4	\\
\\
J07285137-3014490	&	0.175	&	143.71	&	2002.83	&	9	\\
	&	0.485	&	169.9	&	2008.86	&	4	\\
\\
J08224744-5726530	&	8.6	&	23	&	1999.99	&	5	\\
	&	8.429	&	26.1	&	2008.88	&	4	\\
\\
J21103147-2710578	&	9.4	&	313	&	1998.59	&	5	\\
	&	9.501	&	313.2	&	2008.87	&	4	\\
\\
J22171899-0848122AB	&	7.8	&	213	&	1998.79	&	5	\\
	&	7.95	&	213.2	&	2008.87	&	4	\\
\\
J22171899-0848122BC	&	0.978	&	305.8	&	2001.60	&	10	\\
	&	0.97	&	316.7	&	2008.87	&	4	\\
\\
J22332264-0936537	&	1.66	&	272.25	&	1997.6	&	6	\\
	&	1.571	&	279.73	&	2005.44	&	9	\\
	&	1.421	&	98.6/278.6	&	2008.87	&	4	\\
\\
J22382974-6522423	&	0.770	&	16	&	1991.25	&	2	\\
	&	0.842	&	155.8	&	2008.87	&	4	\\
\\
J22401867-4931045	&	4.2	&	40	&	1999.72	&	5	\\
	&	4.039	&	41.0	&	2008.87	&	4	\\
\\
J23581366-1724338	&	1.904	&	355.3	&	2005.54	&	9	\\
	&	1.989	&	355.7	&	2008.87	&	4	\\
\par
\hline
\par
\end{tabular}
\tablebib{(1)~\citet{Dommanget2002}; (2) \citet{Perryman1997}; (3)  \citet{StrigachevLampens2004}; (4) This work;
(5) \citet{Cutri2003}; (6) \citet{McCarthy2001}; (7)  \citet{Kasper2007}; (8) \citet{Mason2001}; (9) \citet{Daemgen2007};
(10) \citet{Beuzit2004}.
}
\end{table}